Author's response to reviews

Title: Routine Invasive Management After Fibrinolysis in Patients With ST-Elevation Myocardial Infarction: A Systematic Review of Randomized Clinical Trials

Authors:

Peter Bogaty (peter.bogaty@fmed.ulaval.ca)
Kristian B Filion (kristian.filion@mail.mcgill.ca)
James M Brophy (james.brophy@mcgill.ca)

Version: 6 Date: 13 June 2011

Author's response to reviews:

June 12, 2011

The BMC Series Editorial Production Team
Journal: BMC Cardiovascular Disorders
MS ID: 1516694379552082

Dear ‘Team’,

We respond to your letter of June 7 regarding this manuscript.

We have complied with all your requests.

On the page that follows, we reproduce all your requests and specifically indicate after each request that we have complied.

We look forward to this publication.

Sincerely,

Peter Bogaty, MD
(On behalf of all authors)

Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs) DONE

Structure: Change 'Introduction' to 'Background'. DONE

References: Reference 46 has been cited as 4646. Please change. DONE

Tables: Please do not use ranges when listing tables. Please list all the figures.
Figures: Please do not use ranges when listing figures. Please list all the figures.

Structure: Please remove the shaded box and include the text within the structure.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Email address: please include email addresses for all the authors on the title page, using the following format:
AB: abcd@institution.ac.uk
EF: efg@generic.co.uk
IJ: IJKL@corporation.com

Funding: Please move the statement of funding to the Acknowledgements section and remove it from the title page.

Disclaimer: Please move the disclaimer to the Acknowledgements section and remove it from the title page.

Tables: please remove the visible vertical lines from your tables.

Appendix: Please remove visible vertical lines.

Figures: The figure file should not include the title (e.g. Figure 1... etc.) or the figure number.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.